# One-to-one volunteer befriending to reduce symptoms of depression in people with intellectual disability: a feasibility RCT

Afia Ali,<sup>1\*</sup> Emma Mckenzie,<sup>2</sup> Zara Rasoul,<sup>1</sup> Angela Hassiotis,<sup>1</sup> Stefan Priebe,<sup>3</sup> Brynmor Lloyd-Evans,<sup>1</sup> Rebecca Jones,<sup>1,4</sup> Monica Panca,<sup>4</sup> Rumana Omar,<sup>5</sup> Sally Finning,<sup>6</sup> Shirley Moore,<sup>7</sup> Christine Roe<sup>7</sup> and Michael King<sup>1,4</sup>

**Declared competing interests of authors:** Several of the authors have held membership positions on National Institute for Health Research funding committee groups: Michael King was a member of the Rapid Trials and Add-on Studies Board (2012–12); Stefan Priebe was a member of the Health Technology Assessment (HTA) Mental, Psychological and Occupational Health (MPOH) Panel (2014–18); Rumana Omar was a member of the HTA General Committee (2016–20); and Angela Hassiotis is currently a member of the HTA Commissioning Committee (2018–present).

Published October 2021 DOI: 10.3310/phr09100

## **Plain English summary**

Volunteer befriending to reduce depression symptoms Public Health Research 2021: Vol. 9: No. 10

DOI: 10.3310/phr09100

NIHR Journals Library www.journalslibrary.nihr.ac.uk

<sup>&</sup>lt;sup>1</sup>Division of Psychiatry, University College London, London, UK

<sup>&</sup>lt;sup>2</sup>Research and Development Office, Goodmayes Hospital, North East London NHS Foundation Trust, London, UK

<sup>&</sup>lt;sup>3</sup>Unit for Social and Community Psychiatry (World Health Organization Collaborating Centre for Mental Health Services Development), Queen Mary University of London, London, UK

<sup>&</sup>lt;sup>4</sup>Priment Clinical Trials Unit, University College London, London, UK

<sup>&</sup>lt;sup>5</sup>UK Department of Statistical Science, University College London, London, UK

<sup>&</sup>lt;sup>6</sup>Outward, London, UK

<sup>&</sup>lt;sup>7</sup>The Befriending Scheme, Sudbury, UK

<sup>\*</sup>Corresponding author afia.ali@ucl.ac.uk

# **Plain English summary**

People with intellectual disability (also known as learning disability) often have few friends and limited access to social support, which can lead to loneliness. This can have a negative effect on their self-esteem and lead to depression. Befriending could help individuals with intellectual disability to access more community activities and emotional support, which could lead to improved well-being, but there have been no trials evaluating befriending in this population.

## **Methods**

We carried out a pilot trial of one-to-one befriending by volunteers in people with intellectual disability who had symptoms of depression. Participants were randomised to the intervention group or control group. Both groups received usual care and a booklet of local activities. We recruited participants from one NHS trust and two befriending schemes that managed the befriending intervention (i.e. recruitment, training, matching and supervision of volunteers). Pairs were asked to meet for at least 1 hour per week, over 6 months. We were aiming to recruit at least 35 participants, to match 70% of those in the intervention group to a volunteer, for less than 30% of participants and volunteers to drop out of the study, for befriending to be considered acceptable (based on feedback from interviews) and for the intervention be delivered as intended (at least 10 meetings between pairs).

#### Results

We recruited only 16 participants and 10 volunteers because of delays in setting up the study and challenges with our recruitment approach. Six of the eight participants in the intervention group were matched to a volunteer. Only two volunteers dropped out of the study. Matched pairs met 12 times and the most frequent activities involved going to cafes and restaurants, and on walks. Experiences of befriending were largely positive.

## **Conclusion**

It will not be possible to carry out a large randomised trial using the same recruitment approach.

## **Public Health Research**

ISSN 2050-4381 (Print)

ISSN 2050-439X (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PHR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/phr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

## Criteria for inclusion in the Public Health Research journal

Reports are published in *Public Health Research* (PHR) if (1) they have resulted from work for the PHR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Public Health Research* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

#### PHR programme

The Public Health Research (PHR) programme, part of the National Institute for Health Research (NIHR), is the leading UK funder of public health research, evaluating public health interventions, providing new knowledge on the benefits, costs, acceptability and wider impacts of non-NHS interventions intended to improve the health of the public and reduce inequalities in health. The scope of the programme is multi-disciplinary and broad, covering a range of interventions that improve public health.

For more information about the PHR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/public-health-research.htm

#### This report

The research reported in this issue of the journal was funded by the PHR programme as project number 16/122/57. The contractual start date was in July 2018. The final report began editorial review in September 2020 and was accepted for publication in May 2021. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PHR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the PHR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PHR programme or the Department of Health and Social Care.

Copyright © 2021 Ali et al. This work was produced by Ali et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

## NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

## **NIHR Journals Library Editors**

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk